Author's response to reviews

Title: An international comparative study of blood pressure in populations of European vs. African descent

Authors:

Richard S Cooper (rcooper@lumc.edu)
Katharina Wolf-Maier (kwolf3@lumc.edu)
Amy Luke (aluke@lumc.edu)
Adebowale Adeyemo (deboadeyemo@scientist.com)
Jose R Banegas (joseramon.banegas@uam.es)
Terrence Forrester (terrence.forrester@uwimona.edu.jm)
Simona Giampaoli (simona.giampaoli@iss.it)
Michel Joffres (michel.joffres@dal.ca)
Mika Kastarinen (Mika.Kastarinen@uku.fi)
Paola Primatesta (p.primatesta@public-health.ucl.ac.uk)
Birgitta Stegmayr (Birgitta.Stegmayr@medicin.umu.se)
Michael Thamm (thammm@rki.de)

Version: 5 Date: 1 December 2004

Author's response to reviews:

No changes to author information.

In the Abstract, first sentence, changed last word from "disease" to "condition".

In a single instance in the text changed "blood pressure" to "BP.

Changed your change of "assorted" (you had used "assimilated", which does not convey the intended sense.

Added a reference, new number 37, to remain up to date, and re-numbered those subsequent.

I don't have the current version of the figures - not attached - and was unsure how you intended to crop the top margins.

Richard Cooper